Author's response to reviews

Title: Transmission of Specific Genotype Streptomycin Resistant Strains of Mycobacterium tuberculosis in the Tokyo Metropolitan Area in Japan

Authors:

Akihiro Ohkado (ohkadoa@jata.or.jp)
Yoshiro Murase (ymurase@jata.or.jp)
Masaaki Mori (hlctryj@yahoo.co.jp)
Naoki Hasegawa (hasegawn@sc.itc.keio.ac.jp)
Goro Otsuka (ootsuka-g@city.kawasaki.jp)
Michiko Nagamine (Michiko_Nagamine@member.metro.tokyo.jp)
Hideo Maeda (Hideo_Maeda@member.metro.tokyo.jp)
Kazuhiro Uchimura (uchimura@jata.or.jp)
Masako Ohmori (ohmori@jata.or.jp)
Norio Yamada (nyamada@jata.or.jp)
Shinji Maeda (maeda@jata.or.jp)
Seiya Kato (kato@jata.or.jp)
Toru Mori (tmori-rit@jata.or.jp)
Nobukatsu Ishikawa (ishikawa@jata.or.jp)

Version: 3 Date: 21 August 2009

Author's response to reviews:

Dear Editors,

We made both major and minor revisions in accordance with your formatting requirements.

One figure as Figure 1 and one table as Table 1 are included in the text after references.

Thank you for your kind assistance.

Akihiro Ohkado